ERF WORKING PAPERS SERIES

On the Determinants and Outcomes of IMF Loans: A Political Economy Approach

Jala Youssef and Chahir Zaki



Working Paper No. 1492 October 2021

ON THE DETERMINANTS AND OUTCOMES OF IMF LOANS: A POLITICAL ECONOMY APPROACH

Jala Youssef¹ and Chahir Zaki²

Working Paper No. 1492

October 2021

Comments from Kabbashi Suliman on an earlier draft are gratefully acknowledged. We also thank participants in the Economic Research Forum (ERF) 27th Annual Conference for useful discussions, especially Ibrahim Elbadawi and Thibault Lemaire.

Send correspondence to: Jala Youssef University of Paris 1 Panthéon-Sorbonne Jala.Emad-Youssef-Erian@etu.univ-paris1.fr

jala.emad@gmail.com. ² Associate Professor of Economics and Director of the French Section, Faculty of Economics and Political Science, Cairo University and Lead Economist, Economic Research Forum (ERF). Email: chahir.zaki@feps.edu.eg

¹ PhD Candidate in Economics, University of Paris 1 Panthéon-Sorbonne (Cotutelle with Cairo University). Email:

First published in 2021 by The Economic Research Forum (ERF) 21 Al-Sad Al-Aaly Street Dokki, Giza Egypt www.erf.org.eg

Copyright © The Economic Research Forum, 2021

All rights reserved. No part of this publication may be reproduced in any form or by any electronic or mechanical means, including information storage and retrieval systems, without permission in writing from the publisher.

The findings, interpretations and conclusions expressed in this publication are entirely those of the author(s) and should not be attributed to the Economic Research Forum, members of its Board of Trustees, or its donors.

Abstract

The main objective in this paper is to empirically analyze the economic and political determinants of IMF lending in low- and middle-income countries. Compared to the existing literature, our main contribution is twofold. First, using the IMF Monitoring of Fund Agreements (MONA) database, we merge domestic political and institutional factors with international political economy factors to analyze IMF lending determinants. Second, we use the predicted values of determinants of IMF lending as instruments to explain the consequences of this lending on economic outcomes. Our main findings show that economic and political proximity to the IMF major shareholders matter for the likelihood of obtaining an IMF non-concessional loan. Furthermore, most of the loans seem to exert either an insignificant or a negative effect on the trend component of GDP, confirming that such loans can stabilize the economies in the short term without improving the long run steady growth. Yet, democratic regimes compared to autocratic ones improve the effects of these loans on economic growth and other outcomes (such as the current account and inflation). By contrast, key physical and human capital variables do not seem to be significantly affected by such loans.

Keywords: IMF lending, Political economy, Structural Adjustment Programs. **JEL Classifications:** P16, F33, F34, F55.

ملخص

الهدف الرئيسي من هذا البحث التحليل التجريبي للمحددات الاقتصادية والسياسية الخاصة بالقروض التي يمنحها صندوق النقد الدولي للبلدان منخفضة ومتوسطة الدخل. مقارنةً بالأدبيات الحالية، فإن مساهمتنا الرئيسية ذات شقين. أولاً، عند والمؤسسية المحلية مع عوامل الاقتصاد السياسي الدولي لتحليل محددات الإقراض من صندوق النقد الدولي. ثانيًا، نستخدم تُظهر المتوقعة لمحددات الإقراض من صندوق المقد الدولي كأدوات لشرح عواقب هذا الإقراض على المخرجات الاقتصادية. تُظهر النتائج الرئيسية المحلية مع عوامل الاقتصاد السياسي الدولي كأدوات لشرح عواقب هذا الإقراض على المخرجات الاقتصادية. والمؤسسية المحلية مع عوامل الاقتصاد السياسي الدولي كأدوات لشرح عواقب هذا الإقراض على المخرجات الاقتصادية. تُظهر النتائج الرئيسية التي توصلنا إليها أن التقارب الاقتصادي والسياسي من المساهمين الرئيسيين في صندوق النقد الدولي يعد مهمًا لاحتمالية الحصول على قرض غير ميسّر من صندوق النقد الدولي وعلاوة على ذلك، يبدو أن لمعظم القروض تأثير غير مهم أو سلبي على مكون اتجاه إجمالي الناتج المحلي، مما يؤكد أن هذه القروض قد تحقق استقرارًا بين الاقتصادات على المدى القصير دون تحسين النمو المطرد على المدى الطويل. ومع ذلك، تعمل الأنظمة الديمقراطية مقارنة بالأنظمة وعلى القصير دون تحسين النمو المطرد على المدى الطويل. ومع ذلك، تعمل الأنظمة الديمقراطية مقارنة بالأنظمة وعلى النقيض من ذلك، يبدو أن المتغيرات الرئيسية لرأس المال المادي والمخرجات الأخرى (مثل الحساب الجاري والتضخم).

1. Introduction

The role and mission of the International Monetary Fund (IMF) have evolved along with the global economy. In addition to economic surveillance and technical assistance, one of the IMF main objectives is to provide financial support to its member countries to address actual or potential balance of payments problems. This suggests that the IMF lending should be mainly based on technical economic considerations. However, this does not seem to be the case and controversial anecdotal evidence along with some studies found that politics largely play a role in the IMF's lending decisions. This is why the IMF faced some serious criticism and calls for its reform took place across the political spectrum. Critics included among others promoting moral hazard and dependency or recidivism through repetitive lending and imposing stabilization reforms that might not correspond to local needs (Steinward and Stone, 2008; Bird, 2007; Bird et al., 2004; Dreher and Vaubel, 2004; Stone, 2004). Yet, less attention has been attributed to the role of institutions in the recipient country. Thus, this paper examines both the economic and political determinants of IMF loans and how the domestic politics of the recipient country affects the outcomes of these loans.

Several theoretical models explain why politics have to be included when analyzing IMF lending. From a public choice theory perspective, the Fund can be considered as an independent actor that aims at maximizing its own objective function incorporating power, prestige, responsibility, and resources (Bird, 2007; Vaubel, 1996 and 1986). If a principal-agent perspective is adopted, the Fund has an interest in providing lending and its principals (major shareholders) would prefer that the Fund enforces conditionality (Stone, 2004 and Vaubel, 1986). On the domestic side, a government can resort to the Fund to overcome domestic opposition to policy reforms (Putnam, 1988 and Vreeland, 2003). On the empirical front, a universal consensus is not achieved yet concerning the determinants and outcomes of IMF lending (Bird, 2007). The literature offered a variety of models explaining IMF programs participation. Early studies which attempted to explain IMF lending by exclusively relying on economic factors suffered from low explanatory power (Bird, 2007; Thacker, 1999). Hence, subsequent literature augmented their models with some political economy aspects. However, results from these models are sometimes contradictory and there is a little consensus on which political determinants really matter (Steiwand and Stone, 2008; Sturm et al., 2005) and this represents an important motivation for this paper.

Against this backdrop, our main objective in this paper is to empirically analyze the economic and political determinants of IMF lending in low- and middle-income countries and show how the domestic politics affects the outcomes of these loans. Compared to the existing literature, our main contribution is twofold. First, using the IMF Monitoring of Fund Agreements (MONA) database, we merge domestic political and institutional factors with international political economy factors to analyze IMF lending determinants. Second, we use the predicted values of determinants of IMF lending as instruments to explain the consequences of this lending on economic outcomes. This will help us consider several issues that arise in the literature on the consequences of IMF programs

including the endogeneity treatment and the selection problem since these programs usually come as a response for an economic crisis.

Our main findings show that economic and political proximity to the IMF major shareholders matter for the likelihood of obtaining an IMF non-concessional loan. Furthermore, most of the loans seem to exert either an insignificant or a negative effect on the trend component of GDP, confirming that such loans can stabilize the economies in the short term without improving the long run steady growth. Yet, democratic regimes compared to autocratic ones improve the effects of these loans on economic growth and other outcomes (such as the current account and inflation). By contrast, key physical and human capital variables do not seem to be significantly affected by such loans.

The paper is organized as follows. Section 2 reviews the literature on the IMF loans determinants and outcomes. Section 3 provides a summary of the stylized facts related to the size and different types of IMF lending while relating to the political economy context by region. Section 4 is dedicated to the methodology and data. Section 5 analyzes the empirical findings. Section 6 concludes and offers some policy recommendations.

2. Literature Review

Political economy determinants that were accounted for in the literature include domestic political factors in the member country asking for lending, foreign policy objectives of the most influential IMF shareholders and IMF bureaucratic considerations as follows. On the domestic demand side, the different stages of an IMF program, including the government's decision to resorting to the IMF and accepting conditionality costs, the negotiation stage, the consideration of programs distributional consequences and implementation, entail political dynamics. As for the IMF supply side, various political factors can affect the decision-making process of lending (Sturm et al., 2005). The weighed voting and lending procedures at the Fund can give a room for political dynamics (Thacker, 1999). To that effect, major shareholders can have strong influence on IMF decisions, and they can be rather inclined to provide lending to some countries in comparison to others. This suggests that a country's political proximity to these shareholders can raise the probability and size of an IMF loan.

While analyzing the economic and political determinants of IMF programs participation is important in its own sake, these determinants results can serve as instruments to understand the consequences of IMF lending on economic outcomes. Indeed, politics and institutions in the domestic economy affect the implementation of an IMF program through several channels. First, in democratic countries, negotiations are generally complicated at both the international and domestic levels in order to reach an agreement. This is chiefly attributed to the presence of different stakeholders (including lobbies, trade unions and various chambers) that have to be part of the domestic negotiations. By contrast, in less democratic regimes, this decision is likely to emanate from a centralized power without lengthy negotiations. Indeed, an autocratic regime can have a smaller incentive to resort to the IMF lending since it can undertake itself unpopular reforms. Another contrasting view suggests that a dictatorship can be less constrained by domestic public opinion and hence, can make easier negotiations with the IMF which increases its likelihood of getting the lending (Sturm et al., 2005; Bird and Rowlands, 2001; Przeworski and Vreeland, 2000). Second, the more complicated the negotiations, the more accountable the executive power will be once the loan is obtained. Clearly, governments that are more accountable will be obliged to improve the macroeconomic outcomes in order to be re-elected. Third, since autocratic regimes are likely to stay more in power, they might have more incentives to implement reforms as they expect to see the effect of their policy choices. Moreover, an autocratic regime can be perceived as more capable of enforcing adjustment policies or an impediment for sustainable development since it can compromise good governance (Bird and Rowlands, 2001). Yet, if the executive power is not likely to stay in power in democracies, reforms can suffer from the so-called "time inconsistency". The latter is a situation where the decision-maker's preferences change over time in such a way that a preference can become inconsistent at another point in time, which reduces the likelihood of reforming and hence macroeconomic outcomes will not improve. This is why the discussion on the effects of IMF lending on economic outcomes would be relevant to this paper.

Existing literature is rather inconclusive as to whether IMF programs affect economic growth or not and in which direction and hence, there is no consensus yet on the impact of IMF lending (Steinward and Stone, 2008; Bird, 2007; Dreher, 2006). Much of the existing literature suggest that participation in IMF programs would significantly reduce economic growth (see for example: Vreeland, 2003; Przeworski and Vreeland, 2000; Conway, 1994; Khan, 1990) whereas some studies found positive or mixed results economic growth effects (see for example Bal Gunduz et al., 2013; Dicks-Mireaux et al., 2000). Some of these studies suggest that the participation in IMF programs can reduce growth over the short run but increase it over the long run. These conflicting results arise from different sources including the differences in the methodologies adopted, the types of programs studied (stand-by arrangements versus structural adjustment facilities), the group of countries included, and the time period considered.

On the methodological front, the literature on IMF programs outcomes has focused on possible statistical methods to correct the selection problem that might arise. In comparison to other strands of IMF literature, this selection problem can be more pronounced in the assessment of IMF programs outcomes and it does not seem to be adequately addressed yet (Steinward and Stone, 2008). Various methods have been considered to deal with this problem, including Heckman estimators (see for ex: Przeworski and Vreeland, 2000; Eichengreen et al., 2008) or instrumental variables (IVs) (See for ex: Marchesi and Sirtori, 2011; Dreher, 2006; Barro and Lee, 2005; Butkiewicz and Yanikkaya, 2005; Easterly, 2005) or the method of matching (See for ex: Bird and Rowlands, 2017; Bal Gunduz et al., 2013) or the generalized method of moments (GMM) estimation (See for ex: Clements et al., 2013; Dreher and Walter, 2010). Stubbs et al. (2018)

provides an elaborate discussion on the advantages and the disadvantages of each of the latter methods as follows. The GMM estimation is less frequently used in the literature on IMF outcomes. Despite its flexibility, this method makes strong assumptions about the data generating process and it also includes a risk of overfitting endogenous variables by introducing too many instruments. As for the matching methods, they help tackling the selection bias that would arise from observables by matching participating and non-participating countries in IMF programs with similar context. However, the selection bias from unobservables is not accounted for in this method. With regards to IVs, previous studies relied on a range of political economy variables like the instruments for IMF participation. Afterwards, the outcome variable is regressed on predicted values of IMF participation and observed values of exogeneous variables. The main challenge related to this method is identifying valid instruments for different outcomes. Finally, unlike the matching methods, Heckman estimators can control for the selection on unobservables. In addition, they are more efficient relative to IVs when the selection variable is dichotomous (like the participation in IMF programs). Heckman estimators major limitations relate to their precision that depend on the variance of the inverse Mills ratio which is determined by the predictive capacity of the first stage probit model, the availability of excludable instruments and the absence of country fixed effects in the first stage probit model (due to the incidental parameter problem).

At the regional level, there are several examples confirming the fact that the domestic political narrative matters for IMF lending. For instance, the MENA countries resorted to the IMF in several incidents, including early 1990s and post uprisings. Nevertheless, the focus was mostly on stabilization policies whereas structural reforms were usually delayed which resulted in lack of markets contestability, lack of equal jobs and severe inequalities (see also Youssef and Zaki, 2019). In fact, several countries in the region performed well in terms of economic growth prior to the Arab Spring. However, the different uprisings waves indicated that the region needs to rethink its economic model and social contract to respond to people's aspirations. The IMF tried to account for these changes since its specific strategies for these countries post uprisings highlighted issues of social and economic inclusion (Mossallem, 2016).

3. Stylized Facts

Annex 1-Table 1 provides an overview on the evolvement of the different types of facilities offered by the Fund and Figure 1 describes this evolution over the period 1992 till 2020. Several conclusions can be drawn as follows.

First, the Stand-By Arrangements (SBA) and the Extended Fund Facility (EFF) are the main nonconcessional facilities of the Fund. In terms of number of arrangements, it seems the relative importance of SBA has been somehow declining over the last decade where the total number of SBA arrangements was 20 over 2011- 2020, compared to 71 and 97 over 2001-2010 and 1992-2000 respectively. This is also confirmed in terms of the size of loans where the total SBA arrangements reached 77022 million SDR over the last decade (2011-2020), down from 168939 million SDR in the prior decade (2001-2010).³ In contrast, the EFF importance increased over the last decade where the total number of EFF arrangements increased to 24 over 2011-2020, up from 3 over 2001-2010. Similarly, the total size of EFF arrangements reached 96541 million SDR in 2011-2020 compared to 20142 million SDR a decade earlier.

Second, SBA and EFF facilities do not cover low-income countries since these countries difficult external indebtedness conditions prevent them from borrowing based on the regular nonconcessional conditions. This is why the Fund developed later some concessional facilities particularly designed for these countries, notably the Structural Adjustment Facility (SAF) in 1986 and the Enhanced Structural Adjustment Facility (ESAF) in 1987 (Bal Gunduz et al., 2013; Barro and Lee, 2005). It seems the effectiveness of these two facilities was questioned for several reasons mainly related to the absence of social considerations. For instance, these programs did not include poverty reduction as an explicit goal. Critics accordingly perceived these programs as promoting short-term stabilization objectives ahead of other important social objectives. Furthermore, there was no formal component to assess the programs impact on poor. Hence, some claims suggested that poverty worsens under these programs. To that effect, the Fund carried out reviews of these programs and replaced them with the Poverty Reduction and Growth Facility (PRGF) in 1999 (Bal Gunduz et al., 2013; IMF Website). However, it seems that the PRGF relative importance is lower than ESAF and SAF in terms of both number of arrangements as well as size. The total number of ESAF and SAF agreements over 1992-2000 was 150 agreements with total size of 16658 million SDR whereas the total number of PRGF arrangements over 1999-2010 was 102 arrangements with total size of 9654 million SDR.⁴

Third, over the period 2002 until 2008, the demand for IMF lending dropped sharply and this seems reflected on both the number of arrangements and their size. This might be because global economic conditions were improving, and countries were repaying their commitments to the Fund. Afterwards, the global economy was hardly hit by the financial crisis in 2008 and this indeed had significant impact on the Fund and debates were reinitiated on its role and legitimacy. The IMF response to this crisis included among other introducing new facilities for concessional lending, namely the Extended Credit Facility (ECF), the Standby Credit Facility (SCF), and the Rapid Credit Facility (RCF). The ECF is in fact the equivalent of EFF for low-income countries. Its relative importance has considerably increased over the last decade (2011-2020), where the total number of ECF arrangements increased to 47 arrangements with a total size of 7680 million SDR, compared to 10 arrangements with a total size of 683 million SDR in the prior decade (2001-2010).⁵

³ These numbers exclude any arrangement that was classified in the IMF MONA dataset under two types of programs to avoid misclassification, namely those classified as: SBA-ESF and SBA-SCF.

⁴ These numbers exclude any arrangement that was classified in the IMF MONA dataset under two types of programs to avoid misclassification, namely those classified as PRGF-EFF.

⁵ These numbers exclude any arrangement that was classified in the IMF MONA dataset under two types of programs to avoid misclassification, namely those classified as ECF-EFF.

Finally, the Fund also offers precautionary facilities, namely the Flexible Credit Line (FCL) and the Precautionary and Liquidity Line (PLL). Over the last decade (2011-2020), the total number of FCL and PLL arrangements was lower than ECF, SBA and EFF (22 FCL and PLL compared 51 ECF, 20 SBA and 24 EFF). However, FCL and PLL seems to be quite significant in terms of size compared to other types of loans and this might be explained by their precautionary characteristics.





1.b Size of arrangements (million SDR



Source: Constructed by the authors using IMF Monitoring of Fund Arrangements (MONA) Database

Notes: (i) Data are compiled in these figures based on the starting year of each arrangement. (ii) IMF loans mentioned here stand for: SBA: Stand-By Arrangement; EFF: Extended Fund Facility; ESAF: Enhanced Structural Adjustment Facility; SAF: Structural Adjustment Facility; PRGF: Poverty Reduction and Growth Facility; ECF: Extended Credit Facility; ESF: Exogenous Shocks Facility; SCF: Standby Credit Facility; FCL: Flexible Credit Line; PLL: Precautionary and Liquidity Line; PCI: Policy Coordination Instrument; PSI: Policy Support Instrument; PCL: Precautionary Credit Line. (iii) These figures exclude any arrangement that was classified in the IMF MONA dataset under two types of programs to avoid misclassification, namely those classified as: ECF-EFF, PRGF-EFF, SBA-ESF and SBA-SCF.

Figure 2 describes the distribution of Fund arrangements by region over three decades in terms of number of arrangements and their size. The following conclusions can be drawn about the relative importance of IMF lending by region and the political economy context.

First, it seems that the Middle East and North Africa (MENA) region is the only region that has an increasing number of loans in the last decade (2011-2020) in comparison to the two previous decades, whereas the number of arrangements declined in other regions. This increase in demand for IMF lending can be explained by the political economy context of post Arab Spring Uprisings transition period.

Second, the IMF lending to Europe and Central Asia has been quite important over 1992-2000 and 2001-2010 in comparison to the most recent decade (2011-2020) in terms of both the number of arrangements and their size. This increase in demand for IMF lending in these countries in the earlier two decades can be related to the 1990s transition process of Central European Countries and the 2008 global financial crisis. For instance, the severe economic crisis in Eastern Europe countries during the early transition period was somehow perceived as a result of the communism era mismanagement and this accordingly induced a sort of a popularity for the adjustment measures proposed by the Fund (Pop-Eleches, 2009). This indeed confirms the fact that domestic political perceptions can have an impact on the demand for the Fund lending.

Third, with the end of cold war and the resolution of Latin American debt crisis in 1989, the international economic and political environment seemed more favorable to the compliance to the Washington Consensus. This was also coupled with a boom in international lending and some reforms in IMF conditionality. These factors altogether had led to an increase in demand for IMF lending in Latin American countries in the 1990s. For instance, political and economic elites understood that they had to pursue the economic reforms suggested by the Fund in order to take advantage from the international lending boom since IMF lending would lead to an increase in outside financing (Pop-Eleches, 2009).

Finally, although significant in terms of number of arrangements, the IMF lending to Sub-Saharan seems to be broadly declining. In particular, the total number of arrangements declined from 129 arrangements (worth 12513.19 million SDR) over 1992-2000 to 77 arrangements (worth 7103.44 million SDR) over 2001-2010. This might be due to the fact that these countries were enjoying

stable economic growth in the latter period and hence they did not need to heavily resort to the Fund resources. In the wake of the financial crisis (2010-2013), the size of IMF lending to this region was declining, yet it picked up again in 2014. This is possibly due to the collapse in oil prices in this latter year that adversely affected commodity dependent countries in the region and thereby increased the demand for IMF lending.

Figure 2: IMF lending by region 2.a Number of arrangements



2.b Percentage of total access by region (period average)



Source: Constructed by the authors using IMF Monitoring of Fund Arrangements (MONA) Database *Notes: (i)* Data are compiled in these figures based on the starting year of each arrangement. (ii) These figures include all types of IMF agreements (even the ones classified under two types of arrangements). (iii) Regions are classified based on the World Bank definitions.

The MENA region presents an interesting example on the interaction between the IMF lending and domestic politics. Table 1 provides a description of the IMF lending to the MENA region in terms of size and types of programs contrasted to their regime types (measured by the Polity Scores). The following conclusions can be drawn. First, apart from regimes considered as in transition or interruption periods, it seems that most of the MENA countries which resorted to the Fund lending over 1992-2020 are classified according to Polity Scores as anocracies or autocracies (see Table 1 notes for a detailed description of Polity Scores regimes classification). This confirms an argument previously mentioned that autocratic regimes can be less constrained by public opinion and competitive elections and hence they can easily negotiate with the IMF which would increase their likelihood of obtaining the loan (Sturm et al., 2005; Bird and Rowlands, 2001; Przeworski and Vreeland, 2000). Second, MENA countries demand for IMF lending increased post Arab Spring uprisings. There is indeed a strong economic justification for this increase in demand for IMF support during this period, including the significant deterioration in these countries' external accounts as well as in their fiscal balances. Yet, from a political perspective, an IMF agreement can also help a government in pushing unpopular policies that can face domestic resistance. For instance, rejecting a policy suggested by the Fund can be more costly for domestic opposition since it can send negative signals to creditors and investors (Przeworski and Vreeland, 2000).

Country	IMF Loan	Start year of the program	Total access (in million SDR)	Polity Score
	ESAF	1994	42.36	-7
Algeria	SBA	1994	457.2	-7
	EFF	1995	1169.28	-3
	SBA	1996	8.245	-6
Djibouti	PRGF	1999	19.082	1
<u></u>	PRGF	2008	22.26	2
	EFF	1993	400	-6
	SBA	1997	271.4	-6
Egypt -	EFF	2016	8596.57	-4
-	SBA	2020	3763.64	na
	SBA	2005	475.36	-66 (interruption period or system missing
- -	SBA	2007	475.36	-66 (interruption period or system missing
Iraq -	SBA	2010	2376.8	-66 (interruption period or system missing
-	SBA	2016	3831	6
	EFF	1994	189.3	-2
-	EFF	1996	238.04	-2
-	EFF	1999	127.88	-2
Jordan	SBA	2002	85.28	-2
-	SBA	2012	1364	-3
-	EFF	2016	514.65	-3
-	EFF	2020	926.37	na
	PLL	2012	4117.4	-4
	PLL	2014	3235.1	-4
Morocco -	PLL	2016	2504	-4
-	PLL	2018	2150.8	-4
Tunicio	SBA	2013	1146	-88 (transition period)
Tunisia -	EFF	2016	1952.25	0
	SBA	1996	132.4	-2
_	ESAF	1997	264.75	-2
_	EFF	1997	79.9	-2
Yemen	ESAF	1998	264.75	-2
_	PRGF	2001	238.8	-2
_	ECF	2010	243.5	-2
_	ECF	2014	365.25	-77 (interregnum period)

Table 1. MENA countries IMF arrangements and political regimes

Source: Compiled by authors based on data from the IMF Monitoring of Fund Arrangements (MONA) Database and Polity5 Political Regime Characteristics and Transitions

Notes: (i) The Polity Score captures a regime authority spectrum on a 21-pont scale ranging from -10 (hereditary monarchy) to +10 (consolidated democracy). The Polity scores can also be converted into regime categories in a suggested three part categorization of "autocracies" (-10 to -6), "anocracies" (-5 to +5 and three special values: -66, -77 and -88), and "democracies" (+6 to +10). (ii) Polity score corresponds to the start year of the program. (iii) IMF loans abbreviations mentioned here stand for: ECF: Extended Credit Facility; EFF: Extended Fund Facility; ESAF: Enhanced Structural Adjustment Facility; SBA: Stand-By Arrangement; PLL: Precautionary and Liquidity Line; PRGF: The Poverty Reduction and Growth Facility

4. Econometric Specification and Data

4.1. Econometric Specification

Following Barro and Lee (2005), we will study the economic and political determinants of IMF lending. Using a large panel of low- and middle-income countries (including MENA countries) over the period 1993-2019⁶, we estimate a panel regression where i is the country and t is the time as follows:

$$Ln(Loan)_{it} = \beta_0 + \beta_1 X_{it} + e_{it} \tag{1}$$

The dependent variable $(Ln(Loan)_{it})$ measures the amount of the IMF credit⁷ obtained by country *i* in year *t*. As it was mentioned before, four types of loans that are taken into consideration: Poverty Reduction and Growth Facility (PRGF), Stand-By Arrangement (SBA), Extended Fund Facility (EFF) and Extended Credit Facility (ECF). Following previous literature (Sturm et al., 2005; Bird and Rowlands, 2001), we include (X_{it}) a vector of time variant economic determinants affecting the IMF credit as follows: international reserves, current account balance as percentage of GDP, inflation rate, exchange rate and government budget deficit as percentage of GDP. These variables are lagged once to avoid potential endogeneity problem and to improve the confidence in assigning causality.

Yet, since not all low- and middle-income countries obtain an IMF loan, equation (1) is likely to suffer from a selection bias. In order to overcome the selection bias, we estimate our regressions using a Heckman selection procedure (Heckman, 1976). Our estimable equation is:

$$Ln(Loan_{it}) = X_{it} \beta + \eta_{it} \tag{2}$$

The dependent variable is not, however, observed for all countries since some of them did not obtain an IMF loan. It is only observed if

$$Z_{it}\gamma + \varepsilon_{it} > 0$$

While the first step (the probability of obtaining an IMF loan) is estimated using a probit model, the second step (the value of the loan) corrects for self-selection by incorporating a transformation of the predicted loan probabilities. The Heckman selection technique helps us therefore to overcome the problem of selection bias. In order to correct this bias, it is better to have (X) as a strict subset of (Z). Hence, our exclusion variables (Z) that explain the likelihood of obtaining an IMF loan is measured by political and economic proximity to the IMF major shareholders in

⁶ We use a group of 156 countries including all low- and middle-income countries according to the World Bank classification and all countries which were enrolled in any IMF agreement over the period of our study. See Annex 2 for a complete list of these countries.

⁷ Some studies exclusively focus on two types of IMF programs (Stand-By Arrangement (SBA), Extended Fund Facility (EFF)) since they are the major programs offered by the IMF.

addition to (X) specified above. Indeed, the two variables *Pol Prox.* and *Econ. Prox* account for proxies for the country's political and economic proximity to the IMF major shareholders which are also permanent members of the UN Security Council (United States, France, China and the United Kingdom): the country's votes in the UN General Assembly along with each major shareholder and the ratio of the country's bilateral trade with each major shareholder to the country's GDP (Barro and Lee, 2005). To construct these two variables, we multiply the share of bilateral trade and the vote similarity by the quota share of the major shareholders in the total IMF quota, then we add these multiplicative terms to obtain a weighted average of economic and political proximity respectively. Finally, y_{it} and ε_{it} are the discrepancy terms.

The third step of our analysis examines the effect of the obtained loans on macroeconomic outcomes. We mainly focus on GDP growth by distinguishing between the cyclical and trend components as measures of short- and long-term effects respectively as follows:

$$Growth_{it} = \alpha_0 Inv_{it} + \alpha_1 School_{it} + \alpha_2 Land_{it} + \alpha_3 Nat. Res_{it} + \alpha_4 Loan_{it} + v_i + \varepsilon_{it}$$
 (3)

Where *Growth*_{*it*} is measured by the growth rate of real GDP, *Inv*_{*it*} gross fixed capital formation, *School*_{*it*} secondary school enrollment, *Land*_{*it*} the share of arable land, *Nat. Res.*_{*it*} the share of natural resources rents to GDP, *Loan*_{*it*} the predicted loan from the Heckman selection model, *v*_{*i*} the fixed effect for each country in order to account for time-invariant unobservable country-specific factors and ε_{it} the discrepancy term. Using the Hodrick-Prescott (HP) filter, our dependent variable is decomposed to two parts: the cyclical component (measuring the output gap) and the trend one (measuring potential GDP). This distinction helps us disentangle the stabilization (short term) vs. the allocation effect (long run) of IMF loans on growth. In order to test our results sensitivity to the GDP growth decomposition methodology, we replace this HP filter with an alternative statistical filtering methodology, the Butterworth (BW) filter, in order to compute the structural and cyclical components of GDP growth. The latter filter is considered as a valid alternative to the HP filter and both are high-pass filters.⁸

Our analysis is extended in two ways. First, we examine how the IMF loan can exert a differential impact on economic growth depending on the regime type (autocratic, anocratic and democratic regimes)⁹. For instance, an autocratic regime can have a smaller incentive to resort to the IMF lending since it can undertake itself unpopular reforms. Another contrasting view suggests that a dictatorship can be less constrained by domestic public opinion and hence, can make easier

⁸ High-pass filters, such as the HP and the BW filters, allow for stochastic cycles meeting a minimum frequency level. Band filters, such as Christiano-Fitzgerald (CF), and Baxter-King (BK) filters, allow only stochastic cycles within a specified range of frequencies with any frequencies outside this range filtered out (Fedderke and Mengisteab, 2017).

 $^{^{9}}$ The regime type classification is based on Polity Scores. The Polity Score captures a regime authority spectrum on a 21-pont scale ranging from -10 (hereditary monarchy) to +10 (consolidated democracy). The Polity scores can also be converted into regime categories in a suggested three part categorization of "autocracies" (-10 to -6), "anocracies" (-5 to +5 and three special values: -66, -77 and -88), and "democracies" (+6 to +10).

negotiations with the IMF which increases its likelihood of getting the lending (Sturm et al., 2005; Bird and Rowlands, 2001; Przeworski and Vreeland, 2000). On the implementation side, depending on the context, an autocratic regime can be perceived as more capable of enforcing adjustment policies or an impediment for sustainable development since it can compromise good governance (Bird and Rowlands, 2001). Second, we examine the association between the different types of loans and various macroeconomic outcomes that include both stabilization variables (fiscal balance, debt services, reserves, current account, inflation and exchange rate) and key physical and human capital variables (investment andschooling). Later, we plan to extend our analysis in several ways. The previous model will be contrasted to another two models where our dependent variable will be the participation rate in IMF loans (F_{it} : fraction of months during each year that a country operated under an IMF loans). We will accordingly use Tobit models with the same previous explanatory variables.

4.2. Data

Studying the IMF programs determinants and impact on economic outcomes is currently more feasible since the IMF made detailed data on its lending programs publicly available. Hence, with regards to the IMF related variables, two complementary sources of data are used: the IMF Monitoring of Fund Arrangements (MONA) Database and the IMF Financial Data Query tool. In particular, the IMF MONA database is available for all IMF historical arrangements since 1993 till present.

The economic determinants affecting the IMF credit (international reserves, current account balance as percentage of GDP, inflation rate, and exchange rate) as well as the growth rate of real GDP and its determinants (human capital (secondary school enrollment), physical investment as a measure of capital, natural resources and arable land as a measure of endowments and the changes in terms of trade) and are obtained from the World Development Indicators of the World Bank. The government budget deficit as percentage of GDP is obtained from the IMF Fiscal Monitor database. The regime type classification is based on the Polity Scores obtained from the Polity5 Political Regime Characteristics and Transitions Database, 1800-2018.

The exclusion variables that explain the likelihood of obtaining an IMF loan are proxies for the country's economic and political proximity to the IMF major shareholders and members of Security Council (United States, France, China and the United Kingdom) as follows: the country's votes in the UN General Assembly along with each major shareholder are obtained from Erik Voeten Dataset for United Nations General Assembly Voting Data (Harvard Dataverse), the ratio of the country's bilateral trade with each major shareholder are obtained from the UN Comtrade dataset.

5. Empirical Findings

5.1. Determinants of Obtaining an IMF Loan

As it was mentioned before, not all low- and middle-income countries obtain an IMF loan. This is why we run a Heckman selection model where economic and political proximity and lagged macroeconomic variables determine the likelihood of obtaining an IMF loan whereas its size is determined by lagged macroeconomic variables. The coefficient of the inverse Mills ratio (Lambda) is positive and significant for both EFF and ECF, thereby suggesting that a positive selection has occurred.

First, for both EFF and SBA, economic and political proximity exert a positive impact on the likelihood of obtaining a loan. It is important to note that these loans are non-concessional, have a longer time span and thus more demanding in their conditionality/rates. These proximity variables are negatively associated to the PRGF that is a concessional loan given to low-income countries. The same result holds for the ECF that replaced the PRGF as the main tool for addressing balance of payments problems for low-income countries (see Table 2).

As per the determinants of the value of the loan, it is clear that the lower the international reserves the higher the EFF. This can be attributed to the fact that this loan is mainly provided to countries that have problems of balance of payments. The same result holds for the PRGF. As per the fiscal stance, the lower the fiscal balance (a larger deficit), the higher the IMF loan (for both ECF and SBA). This shows how such loans help consolidate the fiscal stances of recipient countries.

	F1	FF	PR	CF	SI	BA	F	CF
	Ln(Loan)	Prob(Loan)	Ln(Loan)	Prob(Loan)	Ln(Loan)	Prob(Loan)	Ln(Loan)	Prob(Loan)
Cur. Acc/GDP(-1)	0.0138	-0.0160*	0.0268**	-0.0309***	0.0992***	0.0183***	-0.106	-0.0331***
	(0.0458)	(0.00902)	(0.0119)	(0.00662)	(0.0201)	(0.00646)	(0.0886)	(0.0103)
Fisc. Bal./GDP(-1)	-0.0192	0.0428**	-0.0178	0.0442***	-0.221***	-0.0269**	-0.280***	-0.000711
	(0.0947)	(0.0172)	(0.0153)	(0.0120)	(0.0391)	(0.0124)	(0.102)	(0.0212)
Res./GDP(-1)	-18.12***	-3.960***	-5.615***	-3.276***	4.630***	-1.501***	-1.448	0.804**
	(6.509)	(1.077)	(1.423)	(0.688)	(1.634)	(0.515)	(2.873)	(0.410)
Ln(Ex. Rate)(-1)	0.236**	0.0421	-0.116***	0.0830***	-0.0676	-0.0544***	0.427	0.116***
	(0.110)	(0.0265)	(0.0358)	(0.0185)	(0.0615)	(0.0182)	(0.285)	(0.0338)
Ln(Inflation)(-1)	0.0113	0.00151	0.0257**	-0.00441	0.00397	0.00729***	-0.00779	0.00181
	(0.0140)	(0.00260)	(0.0102)	(0.00490)	(0.00407)	(0.00259)	(0.0504)	(0.00411)
Lambda (Mills ratio								
coefficient)	3.339578**		0.0177608		0.4764441		3.481444*	
	(1.613644)		(0.2368194)		(0.5068588)		(1.967045)	
Proxim. Vote		39.74***		-68.91***		54.51***		-30.29*
		(12.09)		(9.685)		(7.687)		(16.22)
Proxim. Trade		0.827		-15.84***		8.217***		-33.46***
		(2.508)		(2.595)		(1.553)		(10.09)
Constant	-0.392	-3.777***	5.108***	2.966***	4.840***	-4.059***	-6.277	-0.552
	(3.207)	(0.724)	(0.407)	(0.538)	(0.705)	(0.453)	(6.038)	(0.960)
Observations	1,390	1,390	1,390	1,390	1,390	1,390	1,390	1,390

rvations1,3901,3901,3901,3901,3901,3901,390Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-
specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.</th>

5.2. The Effect of IMF Loans on Growth Components

After estimating the determinants of IMF loans, we use the predicted value of loans in a growth equation. Table 3 present the results of the four types of loans for three dependent variables: the cyclical component, trend component and overall growth.

Interestingly, the SBA loan variable does not have a significant effect on the trend component of GDP neither on the cyclical one. As per EFF, PRGF and ECF, IMF loans have a negative effect on the trend component and a positive or an insignificant one on the cyclical one. This finding confirms the fact that, in some cases, most of the loans target stabilizing the economy (by reducing the gap between observed and potential GDP) without changing its structure and might even deteriorate its long-term steady state (because of a lower trend GDP component). This is result is in line with the one of Easterly (2006) who explains the dependency of the recipient countries on the IMF loans without an improvement of their macroeconomic outcomes in the long run.

As per our control variables, generally, while natural resources rents (that are a source of foreign currency) and investment (that measures physical capital) exert a positive effect on all growth components, arable land effect is mostly insignificant and schooling enrollment has a counter-intuitive negative effect on the trend component of GDP.

To test the sensitivity of our results to the GDP growth decomposition methodology, we used an alternative for the HP filter, the BW filter. Relevant results are reported in Annex 3. Our findings from both filters are broadly consistent.

Tables 4-6 extend the analysis by controlling for the regime type and its interaction with the IMF loan on growth and its different components. While the effect of autocratic regimes is mostly not significant on the three measures of growth, its interaction with the EFF has a negative impact on the trend component of GDP. This is chiefly attributed to the fact, in autocratic regimes, there is less accountability and transparency, which amplifies the negative effect of these loans on the long run economic growth. This interaction is insignificant for PRGF and ECF and positive for SBA (see Table 4). Similarly, when the loan variable is interacted with anocratic regimes (regimes that lie between autocratic and democratic ones), both concessional loans, PRGF and ECF, have a negative impact on the trend component. Such an effect can be explained by the uncertainty that characterizes these regimes (since they are neither purely democratic nor autocratic). By contrast, this interaction turns to be positive for the cyclical component of GDP in the cases of PRGF and SBA. As per the EFF and SBA, the interaction term is not significant as it is shown in Table 5. Finally, Table 6 shows the case of democracies. In the case of EFF and ECF loans, a democracy increases the positive effect of the loan on the trend component of GDP. One potential explanation behind this result is democratic accountability. Indeed, the latter is perceived as a justification for the uses of power (or the use of loans) which leads to the good governance of the loan, enables the concept of checks and balance and allows the public control over the use of public resources (which is the case of the loan). The other interaction terms are generally less significant.

		EFF			PRGF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.122***	0.0432***	0.166***	0.123***	0.0405***	0.164***
	(0.0195)	(0.00433)	(0.0198)	(0.0196)	(0.00431)	(0.0199)
Schooling	0.00682	-0.00842***	-0.00160	0.00267	-0.0108***	-0.00812
	(0.0112)	(0.00250)	(0.0114)	(0.0111)	(0.00244)	(0.0113)
Nat. Rent	0.241***	0.0215***	0.263***	0.228***	0.0219***	0.250***
	(0.0350)	(0.00779)	(0.0355)	(0.0347)	(0.00762)	(0.0352)
Arab. Land	-0.0934*	0.0471***	-0.0463	-0.0874*	0.0499***	-0.0375
	(0.0511)	(0.0114)	(0.0519)	(0.0511)	(0.0112)	(0.0519)
Loan	0.322***	-0.0562***	0.266***	0.568***	-0.222***	0.346**
	(0.0829)	(0.0185)	(0.0841)	(0.171)	(0.0376)	(0.174)
Constant	-2.301*	2.413***	0.112	-5.045***	3.614***	-1.431
	(1.292)	(0.288)	(1.312)	(1.634)	(0.359)	(1.660)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.075	0.114	0.098	0.071	0.133	0.093
Number of countries	95	95	95	95	95	95
		SBA			ECF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.112***	0.0456***	0.158***	0.113***	0.0454***	0.158***
	(0.0202)	(0.00450)	(0.0205)	(0.0193)	(0.00430)	(0.0196)
Schooling	-0.0117	-0.00546**	-0.0172	-0.00929	-0.00468**	-0.0140
	(0.0105)	(0.00234)	(0.0107)	(0.0103)	(0.00228)	(0.0104)
Nat. Rent	0.219***	0.0255***	0.245***	0.202***	0.0216***	0.224***
	(0.0347)	(0.00772)	(0.0352)	(0.0354)	(0.00786)	(0.0358)
Arab. Land	-0.0693	0.0434***	-0.0259	-0.0682	0.0432***	-0.0249
	(0.0513)	(0.0114)	(0.0520)	(0.0509)	(0.0113)	(0.0516)
Loan	0.00552	0.0125	0.0180	-0.206**	-0.0463**	-0.252***
	(0.120)	(0.0266)	(0.121)	(0.0857)	(0.0190)	(0.0868)
Constant	-1.729	2.233***	0.504	-2.650**	2.093***	-0.556
	(1.474)	(0.328)	(1.494)	(1.348)	(0.299)	(1.364)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.062	0.107	0.090	0.067	0.111	0.097
Number of countries	95	95	95	95	95	95

Table 3. Effect of IMF Loans on Growth Components

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.

		EFF			PRGF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.124***	0.0432***	0.167***	0.126***	0.0408***	0.167***
	(0.0195)	(0.00434)	(0.0198)	(0.0196)	(0.00432)	(0.0199)
Schooling	0.00827	-0.00820***	7.09e-05	0.00763	-0.0108***	-0.00317
	(0.0113)	(0.00251)	(0.0115)	(0.0112)	(0.00248)	(0.0114)
Nat. Rent	0.244***	0.0203***	0.265***	0.233***	0.0220***	0.255***
	(0.0352)	(0.00784)	(0.0358)	(0.0347)	(0.00764)	(0.0352)
Arab. Land	-0.0921*	0.0467***	-0.0454	-0.0907*	0.0504***	-0.0403
	(0.0511)	(0.0114)	(0.0519)	(0.0510)	(0.0113)	(0.0519)
Loan	0.320***	-0.0510***	0.269***	0.728***	-0.234***	0.494***
	(0.0838)	(0.0186)	(0.0850)	(0.186)	(0.0410)	(0.189)
Autocracy	1.311	0.0869	1.397	5.385**	-0.0906	5.295**
-	(1.004)	(0.223)	(1.019)	(2.153)	(0.474)	(2.188)
Loan*Autocracy	0.0815	-0.136*	-0.0544	-0.960**	0.0733	-0.886**
	(0.331)	(0.0736)	(0.336)	(0.443)	(0.0976)	(0.450)
Constant	-2.560*	2.402***	-0.158	-6.165***	3.638***	-2.527
	(1.308)	(0.291)	(1.328)	(1.691)	(0.373)	(1.719)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.076	0.117	0.100	0.077	0.134	0.098
Number of countries	95	95	95	95	95	95
		SBA			ECF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.114***	0.0457***	0.160***	0.115***	0.0456***	0.161***
	(0.0203)	(0.00451)	(0.0206)	(0.0193)	(0.00431)	(0.0196)
Schooling	-0.0105	-0.00528**	-0.0158	-0.00819	-0.00448*	-0.0127
-	(0.0106)	(0.00235)	(0.0107)	(0.0103)	(0.00229)	(0.0104)
Nat. Rent	0.221***	0.0256***	0.247***	0.215***	0.0231***	0.238***
	(0.0348)	(0.00772)	(0.0352)	(0.0356)	(0.00792)	(0.0360)
Arab. Land	-0.0675	0.0434***	-0.0241	-0.0636	0.0438***	-0.0199
	(0.0513)	(0.0114)	(0.0520)	(0.0508)	(0.0113)	(0.0514)
Loan	0.0289	0.00463	0.0336	-0.264***	-0.0517***	-0.315***
	(0.122)	(0.0271)	(0.124)	(0.0892)	(0.0199)	(0.0902)
Autocracy	2.851	-0.880	1.971	4.336***	0.537	4.873***
	(2.972)	(0.660)	(3.011)	(1.571)	(0.350)	(1.588)
Loan*Autocracy	-0.301	0.204*	-0.0968	0.770***	0.0809	0.851***
	(0.515)	(0.114)	(0.522)	(0.291)	(0.0648)	(0.294)
Constant	-2.125	2.249***	0.125	-3.210**	2.021***	-1.189
	(1.503)	(0.334)	(1.523)	(1.361)	(0.303)	(1.376)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.064	0.110	0.092	0.074	0.113	0.105
Number of countries	95	95	95	95	95	95

Table 4. Effect of IMF Loans on Growth Components – Autocracy

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.

		EFF			PRGF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.123***	0.0430***	0.166***	0.124***	0.0406***	0.164***
mvestment	(0.0195)	(0.00434)	(0.0198)	(0.0196)	(0.00431)	(0.0199)
Sahaalina	0.00721	-0.00873***	-0.00152	0.00360	-0.0108***	-0.00722
Schooling	(0.00721) (0.0113)		(0.00132)	(0.00360) (0.0111)		(0.0113)
Nat. Rent	(0.0113) 0.241***	(0.00252)	(0.0113) 0.262***	0.225***	(0.00244)	0.248***
Inal. Kelli		0.0214***			0.0225***	
A 1 T 1	(0.0350)	(0.00779)	(0.0355)	(0.0346)	(0.00762)	(0.0352)
Arab. Land	-0.0989*	0.0462***	-0.0527	-0.0856*	0.0485***	-0.0371
_	(0.0513)	(0.0114)	(0.0521)	(0.0513)	(0.0113)	(0.0522)
Loan	0.343***	-0.0495**	0.294***	0.438**	-0.182***	0.256
	(0.0876)	(0.0195)	(0.0889)	(0.201)	(0.0442)	(0.204)
Anocracy	-0.681	-0.0723	-0.754	-2.766**	0.574*	-2.193
	(0.494)	(0.110)	(0.501)	(1.345)	(0.296)	(1.368)
Loan*Anocracy	-0.0486	-0.0363	-0.0848	0.486*	-0.121*	0.365
	(0.150)	(0.0334)	(0.152)	(0.293)	(0.0645)	(0.298)
Constant	-2.052	2.468***	0.417	-4.422***	3.456***	-0.966
	(1.317)	(0.293)	(1.337)	(1.677)	(0.369)	(1.706)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.076	0.115	0.100	0.076	0.136	0.096
Number of countries	95	95	95	95	95	95
		SBA			ECF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.119***	0.0456***	0.164***	0.113***	0.0457***	0.159***
	(0.0201)	(0.00451)	(0.0203)	(0.0194)	(0.00429)	(0.0196)
Schooling	-0.0116	-0.00544**	-0.0170	-0.00911	-0.00472**	-0.0138
-	(0.0104)	(0.00234)	(0.0106)	(0.0103)	(0.00228)	(0.0104)
Nat. Rent	0.221***	0.0254***	0.247***	0.203***	0.0188**	0.222***
	(0.0343)	(0.00773)	(0.0348)	(0.0356)	(0.00788)	(0.0360)
Arab. Land	-0.0809	0.0433***	-0.0376	-0.0703	0.0424***	-0.0280
	(0.0508)	(0.0114)	(0.0515)	(0.0510)	(0.0113)	(0.0516)
Loan	-0.344**	0.0153	-0.329**	-0.247**	-0.0112	-0.258**
	(0.135)	(0.0305)	(0.137)	(0.101)	(0.0223)	(0.102)
Anocracy	-7.364***	0.0347	-7.329***	-0.0486	-0.456***	-0.505
	(1.391)	(0.313)	(1.410)	(0.740)	(0.164)	(0.749)
Loan*Anocracy	1.211***	-0.0117	1.200***	0.101	-0.0981***	0.00248
Lean rineeracy	(0.232)	(0.0522)	(0.235)	(0.147)	(0.0325)	(0.148)
Constant	0.388	2.228***	2.617*	-2.686*	2.276***	-0.410
Constant	(1.514)	(0.341)	(1.535)	(1.377)	(0.305)	(1.393)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.085	0.107	0.111	0.069	0.118	0.098
-	0.083 95	95	95			
Number of countries	90	73	73	95	95	95

 Table 5. Effect of IMF Loans on Growth Components – Anocracy

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p < 0.01, ** p < 0.05, * p < 0.1.

		EFF			PRGF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.122***	0.0427***	0.165***	0.123***	0.0407***	0.164***
	(0.0195)	(0.00434)	(0.0198)	(0.0196)	(0.00431)	(0.0200)
Schooling	0.00688	-0.00896***	-0.00208	0.00273	-0.0105***	-0.00778
	(0.0113)	(0.00252)	(0.0115)	(0.0112)	(0.00245)	(0.0113)
Nat. Rent	0.241***	0.0207***	0.261***	0.227***	0.0227***	0.249**
	(0.0351)	(0.00781)	(0.0356)	(0.0348)	(0.00764)	(0.0353)
Arab. Land	-0.0964*	0.0460***	-0.0504	-0.0901*	0.0491***	-0.0409
	(0.0514)	(0.0114)	(0.0522)	(0.0515)	(0.0113)	(0.0524)
Loan	0.329**	-0.0994***	0.229	0.617***	-0.266***	0.352
	(0.139)	(0.0310)	(0.141)	(0.220)	(0.0484)	(0.224)
Democracy	0.361	0.0565	0.417	0.621	-0.433	0.188
	(0.477)	(0.106)	(0.484)	(1.287)	(0.283)	(1.308)
Loan*Democracy	0.000952	0.0542*	0.0551	-0.0483	0.0798	0.0315
	(0.142)	(0.0317)	(0.144)	(0.284)	(0.0624)	(0.288)
Constant	-2.482*	2.438***	-0.0449	-5.492***	3.854***	-1.638
	(1.309)	(0.291)	(1.330)	(1.769)	(0.389)	(1.799)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.075	0.116	0.099	0.072	0.135	0.094
Number of countries	95	95	95	95	95	95
		SBA			ECF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.116***	0.0458***	0.161***	0.112***	0.0455***	0.158***
	(0.0201)	(0.00451)	(0.0204)	(0.0193)	(0.00429)	(0.0196
Schooling	-0.0124	-0.00546**	-0.0179*	-0.00938	-0.00472**	-0.0141
-	(0.0104)	(0.00234)	(0.0106)	(0.0103)	(0.00228)	(0.0104)
Nat. Rent	0.220***	0.0256***	0.245***	0.212***	0.0185**	0.231**
	(0.0345)	(0.00773)	(0.0349)	(0.0359)	(0.00796)	(0.0363)
Arab. Land	-0.0796	0.0433***	-0.0363	-0.0670	0.0424***	-0.0246
	(0.0510)	(0.0114)	(0.0517)	(0.0510)	(0.0113)	(0.0516
Loan	0.696***	0.0297	0.726***	-0.0551	-0.0877***	-0.143
	(0.191)	(0.0429)	(0.194)	(0.117)	(0.0259)	(0.118)
Democracy	6.044***	0.131	6.174***	-0.892	0.320**	-0.571
·	(1.325)	(0.297)	(1.342)	(0.724)	(0.161)	(0.733)
Loan*Democracy	-1.038***	-0.0253	-1.063***	-0.281**	0.0748**	-0.207
	(0.224)	(0.0502)	(0.227)	(0.143)	(0.0318)	(0.145)
Constant	-5.614***	2.144***	-3.470**	-2.254	1.943***	-0.311
	(1.683)	(0.377)	(1.705)	(1.393)	(0.309)	(1.411)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.080	0.107	0.108	0.071	0.115	0.099
Number of countries	95	95	95	95	95	95

Table 6. Effect of IMF Loans on Growth Components – Democracy

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p < 0.01, ** p < 0.05, * p < 0.1.

5.3. The Effect of IMF Loans on Macroeconomic Outcomes

We extend our analysis by examining the association between different macroeconomic outcomes and different types of IMF loans. These macroeconomic outcomes include both stabilization variables (fiscal balance, reserves, current account, debt services, inflation and exchange rate) and key physical and human capital variables (investment and schooling). We control also for the lagged economic growth. In this section, we limit our analysis to the two extremes, namely autocracies and democracies, since, from an institutional perspective, they are more interesting.

While schooling is positively associated with the SBA loan, the relationship between investment and this loan is insignificant. As per stabilization-related variables, inflation declines, fiscal stance improves, reserves increase, and the exchange rate depreciates. Hence, our results confirm the stabilization effect of the SBA loan. Our interaction variables (whether between SBA and democracy or SBA and autocracy) do not show a stable relationship since some of them are either positive, negative or insignificant (see Table 7).

Moving to the PRGF, our results point out three main issues. First, while key physical and human capital variables are not affected by these loans, some of the stabilization-related ones are. Indeed, the schooling coefficient is insignificant, the investment one is negative and the debt service one is positive (see Table 8). When democracy is interacted with the PRGF loan, some macroeconomic outcomes improve such as current account that improves and inflation that declines. However, the interaction terms between autocracy and the PRGF are all insignificant.

Third, Table 9 shows the results of the EFF. Interestingly, since the objective of the latter is to provide assistance to countries experiencing balance of payments problems, international reserves are positively associated to these loans and exchange rate tend to depreciate. This is intuitive since most of the countries that obtain such loans tend to experience significant devaluations before or during the loan negotiations (as one of the prior actions needed to obtain the loan). Similarly, key physical and human capital variables (schooling and investment) are not positively affected by the EFF. When the regime type is taken into consideration, and similar to the case of PRGF, while current account improves and inflation declines in democratic regimes, international reserves decrease, and the exchange rate increases in autocratic ones. Finally, the results of the ECF are generally similar to the ECF ones with investment and schooling not being affected by the loan, current account balance improves in democratic regimes (see Table 10).

				General Out	comes			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.206***	0.251***	0.0772	0.105***	-0.000321	0.0243***	-1.953***	-0.168
	(0.0437)	(0.0397)	(0.0821)	(0.0318)	(0.000512)	(0.00802)	(0.707)	(0.112)
SBA	0.0415	0.181	4.430***	-0.290**	0.0328***	0.235***	-7.607*	0.947
	(0.248)	(0.227)	(0.414)	(0.143)	(0.00297)	(0.0460)	(4.039)	(0.574)
Constant	-3.134**	20.38***	53.95***	-1.076	-0.0683***	1.202***	68.96***	15.88***
	(1.492)	(1.384)	(2.511)	(0.893)	(0.0178)	(0.273)	(23.99)	(3.452)
Observations	404	399	299	298	415	417	387	293
R-squared Number of	0.066	0.109	0.333	0.072	0.275	0.083	0.031	0.023
countries	71	70	65	56	73	75	69	54
				Democra	lcv			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.210***	0.253***	0.0543	0.106***	-0.000293	0.0252***	-1.928***	-0.162
	(0.0438)	(0.0394)	(0.0809)	(0.0320)	(0.000509)	(0.00803)	(0.704)	(0.112)
SBA	-0.594	0.964**	4.186***	-0.211	0.0419***	0.332***	-17.56**	1.654*
	(0.447)	(0.388)	(0.758)	(0.345)	(0.00508)	(0.0808)	(7.042)	(0.892)
Democracy	-6.113	8.489***	6.629	0.420	0.0996**	1.086	-114.7**	6.966
	(3.743)	(3.154)	(5.793)	(2.860)	(0.0413)	(0.664)	(56.24)	(8.018)
Demo*SBA	0.855*	-1.092**	0.111	-0.102	-0.0127**	-0.133	13.75*	-1.119
	(0.501)	(0.434)	(0.795)	(0.360)	(0.00569)	(0.0904)	(7.795)	(1.053)
Constant	1.364	14.39***	49.59***	-1.391	-0.139***	0.426	152.5***	11.59*
	(3.127)	(2.617)	(5.139)	(2.612)	(0.0342)	(0.549)	(47.70)	(6.264)
Observations	404	399	299	298	415	417	387	293
R-squared	0.074	0.129	0.371	0.074	0.287	0.090	0.044	0.029
Number of								
countries	71	70	65	56	73	75	69	54
				Autocra	•			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.205***	0.239***	0.0717	0.105***	-0.000354	0.0199**	-1.578**	-0.188*
	(0.0442)	(0.0396)	(0.0842)	(0.0318)	(0.000517)	(0.00787)	(0.689)	(0.114)
SBA	0.0494	0.0799	4.444***	-0.275*	0.0325***	0.210***	-6.811*	0.840
	(0.252)	(0.227)	(0.427)	(0.144)	(0.00301)	(0.0453)	(3.961)	(0.587)
Autocracy	1.798	(0.227) -24.18***	(0.427) 4.974	-1,415	-0.0711	(0.04 <i>33)</i> -5.979***	207.0	-17.10
Autocracy				-				
Auto*SBA	(9.488)	(7.999) 4.115***	(13.40)	(1,263)	(0.106)	(1.590)	(141.4) -61.33***	(18.01)
Auto SDA	-0.262		-0.322	230.6	0.0121	1.179***		3.166
Constant	(1.594) -3.190**	(1.393) 21.09***	(2.296) 53.81***	(205.7) -0.00768	(0.0185) -0.0663***	(0.277) 1.340***	(23.66) 67.33***	(3.028) 16.52***
Constant								
Observet	(1.523)	(1.391)	(2.592)	(1.350)	(0.0181)	(0.270)	(23.60)	(3.547)
Observations	404	399	299	298	415	417	387	293
R-squared Number of	0.066	0.133	0.335	0.078	0.276	0.136	0.100	0.028
countries	71	70	65	56	73	75	69	54

Table 7. Effect of SBA on Macroeconomic Outcomes

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p < 0.01, ** p < 0.05, * p < 0.1.

	lect of PKGF			General outco	omes			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0758	0.0988	0.223***	0.0535	0.000586	0.00283	-0.00311	-0.352*
()	(0.102)	(0.0912)	(0.0856)	(0.0749)	(0.000631)	(0.00303)	(0.0882)	(0.197)
PRGF	0.937	-2.641***	-0.536	-2.877***	-0.0512***	0.0569**	-0.549	3.568**
	(0.720)	(0.741)	(0.752)	(0.638)	(0.00587)	(0.0260)	(0.753)	(1.437)
Constant	-10.30***	31.38***	47.81***	10.19***	0.298***	4.528***	9.286***	-0.949
	(2.936)	(3.028)	(2.895)	(2.629)	(0.0235)	(0.104)	(3.003)	(5.755)
Observations	323	337	254	321	312	373	356	344
R-squared	0.009	0.048	0.036	0.073	0.229	0.017	0.002	0.033
Number of countries	50	49	49	48	47	55	54	53
				Democrac	<i>y</i>			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0985	0.0993	0.204**	0.0632	0.000380	0.00272	0.00724	-0.236
	(0.103)	(0.0917)	(0.0861)	(0.0744)	(0.000630)	(0.00306)	(0.0888)	(0.192)
PRGF	0.396	-2.309***	-0.706	-2.228***	-0.0507***	0.0509*	-0.0483	3.553**
	(0.774)	(0.800)	(0.840)	(0.682)	(0.00590)	(0.0278)	(0.802)	(1.488)
Democracy	-8.859*	6.544	-0.753	11.15**	-0.0156	-0.112	9.423*	-2.765
	(4.888)	(5.692)	(4.902)	(4.311)	(0.0549)	(0.185)	(5.347)	(9.726)
Demo*PRGF	2.199**	-1.337	0.718	-2.133**	0.00919	0.0257	-2.174*	-1.975
	(1.116)	(1.362)	(1.167)	(0.983)	(0.0121)	(0.0423)	(1.213)	(2.220)
Constant	-7.763**	29.62***	47.60***	6.491**	0.289***	4.558***	6.726**	3.270
	(3.342)	(3.337)	(3.394)	(2.910)	(0.0249)	(0.117)	(3.371)	(6.309)
Observations	323	337	254	321	312	373	356	344
R-squared	0.023	0.053	0.052	0.101	0.249	0.018	0.012	0.105
Number of countries	50	49	49	48	47	55	54	53
				Autocracy	V			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0756	0.0957	0.223**	0.0538	0.000527	0.00291	-0.00171	-0.353*
	(0.102)	(0.0915)	(0.0861)	(0.0752)	(0.000627)	(0.00304)	(0.0886)	(0.197)
PRGF	0.937	-2.714***	-0.536	-2.877***	-0.0531***	0.0580**	-0.513	3.567**
	(0.723)	(0.749)	(0.765)	(0.641)	(0.00589)	(0.0261)	(0.762)	(1.441)
Autocracy	2.764	-4.551	-0.444	1.798	-0.0874*	0.0933	2.150	-6.683
	(12.88)	(6.509)	(6.424)	(11.11)	(0.0453)	(0.274)	(6.581)	(26.57)
Auto*PRGF	-0.394	0.709	0.248	-0.375	0.0115	-0.00563	-0.330	1.519
	(2.345)	(1.405)	(1.613)	(2.024)	(0.00977)	(0.0557)	(1.421)	(4.840)
Constant	-10.35***	31.77***	47.79***	10.19***	0.308***	4.519***	9.101***	-0.958
	(2.953)	(3.080)	(2.953)	(2.644)	(0.0237)	(0.105)	(3.056)	(5.787)
Observations	323	337	254	321	312	373	356	344
R-squared	0.009	0.050	0.036	0.073	0.245	0.018	0.002	0.033
Number of countries	50	49	49	48	47	55	54	53

Table 8. Effect of PRGF on Macroeconomic Outcomes

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-

specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.

				General outco	omes			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0685	0.213***	-0.0730	0.215**	-0.000136	0.0100	-0.575***	-0.231
	(0.0848)	(0.0573)	(0.136)	(0.0842)	(0.000818)	(0.00618)	(0.206)	(0.261)
EFF	-1.420***	-1.400***	1.365	-0.259	0.0209***	0.456***	0.257	0.826
	(0.517)	(0.360)	(0.984)	(0.532)	(0.00512)	(0.0365)	(1.094)	(1.505)
Constant	5.521	30.67***	72.62***	-1.302	-0.0264	-0.0518	7.490	20.06**
	(3.390)	(2.345)	(6.416)	(3.605)	(0.0333)	(0.228)	(7.102)	(9.723)
Observations	178	189	139	135	189	182	176	149
R-squared	0.053	0.180	0.022	0.067	0.102	0.521	0.056	0.010
Number of countries	35	39	35	31	38	36	37	28
				Democrac	у			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.106	0.215***	-0.0451	0.208**	-5.95e-05	0.00971	-0.470**	-0.191
	(0.0829)	(0.0580)	(0.136)	(0.0867)	(0.000828)	(0.00624)	(0.202)	(0.265)
EFF	-2.276***	-1.511***	0.682	-0.626	0.0208***	0.436***	1.812	1.177
	(0.604)	(0.420)	(1.117)	(0.784)	(0.00630)	(0.0426)	(1.210)	(1.817)
Democracy	-11.95**	-2.502	-4.782	-4.035	-0.0104	-0.474	27.16**	2.854
-	(5.641)	(4.078)	(11.97)	(6.257)	(0.0609)	(0.400)	(11.57)	(17.67)
Demo*EFF	2.019***	0.185	1.421	0.502	-0.000477	0.0520	-4.620***	-0.914
	(0.772)	(0.571)	(1.788)	(0.807)	(0.00818)	(0.0556)	(1.634)	(2.384)
Constant	10.15**	32.17***	73.89***	1.684	-0.0176	0.160	-0.731	19.51
	(4.168)	(2.898)	(7.145)	(5.869)	(0.0437)	(0.284)	(8.249)	(12.35)
Observations	178	189	139	135	189	182	176	149
R-squared	0.123	0.190	0.056	0.071	0.109	0.529	0.131	0.020
Number of countries	35	39	35	31	38	36	37	28
				Autocracy	,			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Lgrowth	-0.0701	0.199***	-0.140	0.219**	0.000394	0.00800	-0.617***	-0.285
	(0.0867)	(0.0583)	(0.117)	(0.0845)	(0.000798)	(0.00619)	(0.206)	(0.265)
lloan_eff2	-1.322**	-1.675***	-6.097***	-0.156	0.0340***	0.441***	-0.975	0.526
	(0.641)	(0.429)	(1.465)	(0.550)	(0.00604)	(0.0436)	(1.311)	(1.863)
authocracy	-79.69	-130.3	-289.0	1.807	4.366***	-30.57**	-274.5	-777.0
	(170.7)	(121.1)	(434.6)	(2.390)	(1.622)	(12.11)	(442.9)	(488.5)
loan_eff_auto2	13.37	21.27	42.52		-0.709***	5.061**	43.87	128.8
	(28.15)	(19.97)	(72.51)		(0.267)	(1.998)	(73.01)	(80.53)
Constant	5.563	33.98***	126.2***	-2.071	-0.164***	0.375	17.43*	31.01**
	(5.099)	(3.458)	(10.57)	(3.753)	(0.0483)	(0.344)	(9.681)	(15.27)
Observations	178	189	139	135	189	182	176	149
R-squared	0.057	0.189	0.296	0.072	0.184	0.543	0.077	0.034
Number of countries	35 Is include a fixed	<u>39</u>	35	31	38	36	37	28

Table 9. Effect of EFF on Macroeconomic Outcomes

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.

				General outco	omes			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0448	0.0496	-0.0585	0.0865***	0.000198	-0.00486*	-0.0575	-0.119*
	(0.133)	(0.0801)	(0.0990)	(0.0295)	(0.000525)	(0.00281)	(0.0678)	(0.0687)
ECF	5.134***	-1.428	2.174	-0.372	0.00182	0.157***	0.965	0.637
	(1.717)	(1.098)	(1.322)	(0.449)	(0.00832)	(0.0411)	(0.819)	(0.889)
Constant	-32.18***	28.93***	41.15***	-1.556	0.181***	4.603***	1.889	4.018
	(7.294)	(5.101)	(5.888)	(2.041)	(0.0347)	(0.178)	(3.569)	(3.801)
Observations	157	156	111	161	131	184	168	162
R-squared	0.066	0.015	0.037	0.064	0.002	0.099	0.015	0.025
Number of countries	27	26	24	27	22	30	29	27
				Democrac	<i>y</i>			
	Current Acc.	Inv.	School.	Fiscal. Balance	Reserves	Ex. Rate	Inflation	Debt Ser.
Growth(-1)	-0.0431	0.0210	-0.0600	0.0814***	-0.000110	-0.00541*	-0.0565	-0.122*
	(0.128)	(0.0795)	(0.0968)	(0.0297)	(0.000508)	(0.00278)	(0.0683)	(0.0691)
ECF	0.136	0.0398	-0.790	-0.487	0.0182	0.0960*	0.918	1.602
	(2.476)	(1.400)	(2.235)	(0.579)	(0.0152)	(0.0551)	(1.119)	(1.338)
Democracy	-30.26**	12.77**	-15.43	0.0132	0.147**	-0.290	-0.460	6.256
	(11.71)	(6.323)	(11.62)	(2.636)	(0.0718)	(0.250)	(5.079)	(6.335)
Demo*ECF	8.312***	-2.324*	4.305	0.174	-0.0247	0.0899	0.0324	-1.375
	(2.782)	(1.402)	(2.636)	(0.585)	(0.0165)	(0.0556)	(1.101)	(1.504)
Constant	-12.16	21.12***	52.88***	-1.431	0.0828	4.826***	2.286	-0.590
	(11.04)	(6.409)	(10.27)	(2.647)	(0.0660)	(0.248)	(5.139)	(5.961)
Observations	157	156	111	161	131	184	168	162
R-squared	0.144	0.065	0.103	0.080	0.109	0.140	0.015	0.033
Number of countries	27	26	24	27	22	30	29	27

Notes: (i) We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors. (ii) Standard errors in parentheses. (iii) *** p<0.01, ** p<0.05, * p<0.1.

6. Conclusion

The IMF lending is supposed to be mainly based on technical economic considerations. However, this does not seem to be the case and controversial anecdotal evidence along with some studies found that politics largely play a role in the IMF's lending decisions. A universal consensus is not achieved yet with regards to the determinants and outcomes of IMF lending. To this effect, the main objective in this paper is to empirically analyze the economic and political determinants of IMF lending in low- and middle-income countries. Compared to the existing literature, our main contribution is twofold. First, using the IMF Monitoring of Fund Agreements (MONA) database, we merge domestic political and institutional factors with international political economy factors to analyze IMF lending determinants. Second, we use the predicted values of determinants of IMF lending as instruments to explain the consequences of this lending on economic outcomes. Our main findings show that economic and political proximity to the IMF major shareholders matter for the likelihood of obtaining an IMF non-concessional loan. Furthermore, most of the loans seem to exert either an insignificant or a negative effect on the trend component of GDP, confirming that such loans can stabilize the economies in the short term without improving the long run steady

growth. Yet, democratic regimes compared to autocratic ones improve the effects of these loans on economic growth and other outcomes (such as the current account and inflation). By contrast, key physical and human capital variables do not seem to be significantly affected by such loans.

At the policy level, the current context of the COVID-19 crisis reinitiates the debate on the role of the IMF given the fact that developing countries, including MENA countries, are more vulnerable to the crisis. As such, this global crisis requires pragmatic solutions with international coordination where the IMF is supposed to play a pivotal role along with domestic stakeholders. Hence, an appropriate evaluation of the determinants of IMF lending and its consequences is timely and important from a policy perspective to investigate how politics and economics affect their outcomes.

Moreover, our results highlight the importance of democratic accountability. Indeed, the latter is perceived as a justification for the uses of power (or the use of loans) which leads to the good governance of the loan, enables the concept of checks and balance and allows the public control over the use of public resources (which is the case of the loan). Obviously, this improves the effect of the loans on macroeconomic outcomes.

References

- Bal Gunduz, Y. et C. Ebeke, B. Hacibedel, L. Kaltani, V. Kehayova, C. Lane, C. Mumssen, N. Mwase, and J.Thornton (2013) "The Economic Impact of IMF Supported Programs in Low Income Countries", *IMF Occasional Paper* No. 13/277
- Barro, RJ. and J-W Lee (2005) "IMF Programs: Who is Chosen and What Are the Effects?", *Journal of Monetary Economics*, Vol. 52, pages 1245-1269.
- Bird, G. and D. Rowlands (2017) "The Effect of IMF Programmes on Economic Growth in Low Income Countries: An Empirical Analysis", *The Journal of Development Studies*, Vol. 53, No. 12, pages 2179-2196.
- Bird, G. (2007) "The IMF: A Bird's Eye View of Its Role and Operations", *Journal of Economic Surveys*, Vol. 21, No. 4, pages 683-745.
- Bird, G., M. Hussain and JP. Joyce (2004) "Many Happy Returns? Recidivism and the IMF", *Journal of International Money and Finance*, Vol. 23, pages 231-251.
- Bird, G. and D. Rowlands (2001) "IMF lending: how is it affected by economic, political and institutional factors?", *The Journal of Policy Reform*, Vol. 4, No. 3, pages 243-270.
- Butkiewicz, J. and H. Yanikkaya (2005) "The Effects of IMF and World Bank Lending on Long-Run Economic Growth: An Empirical Analysis", *World Development*, Vol. 33, No. 3, pages 371-391.
- Clements, B., S. Gupta and M. Nozaki (2013) "What Happens to Social Spending in IMF Supported Programmes?", *Applied Economics*, Vol. 45, Issue 28, pages 4022-4033.
- Conway, P. (1994) "IMF lending programs: participation and impact", *Journal of Development Economics*, Vo. 45, pages 365–391.
- Dicks-Mireaux, L., M. Mecagni and S. Schadler (2000) "Evaluating the effect of IMF lending to low-income countries", *Journal of Development Economics*, Vol. 62, pages 495–526.
- Dreher, A. and S. Walter (2010) "Does the IMF Help or Hurt? The Effect of IMF Programs on the Likelihood and Outcome of Currency Crises", *World Development*, Vol. 38, No. 1, pages 1-18.
- Dreher, A. (2006) "IMF and Economic Growth: The Effects of Programs, Loans, and Compliance with Conditionality", *World Development*, Vol. 34, No. 5, pages 769-788.
- Dreher, A. and R. Vaubel (2004) "Do IMF and IBRD Cause Moral Hazard and Political Business Cycles? Evidence from Panel Data", *Open Economies Review*, Vol. 15, pages 5-22.
- Easterly, W. (2005) "What Did Structural Adjustment Adjust? The Association of Policies and Growth with Repeated IMF and World Bank Adjustment Loans", *Journal of Development Economics*, Vol. 76, Issue 1, pages 1-22.
- Eichengreen, B., P. Gupta and A. Moddy (2008) "Sudden Stops and IMF-Supported Programs", in S. Edwards and MGP. Garcia (Eds.) "Financial Markets Volatility and Performance in Emerging Markets", University of Chicago Press.
- Fedderke, JW and DK. Mengisteab (2017) "Estimating South Africa's Output Gap and Potential Growth Rate", *South African Journal of Economics*, Vol. 65, Issue 2.

- Harrigan, J., C. Wang and H. El-Said (2006) "The Economic and Political of IMF and World Bank Lending in the Middle East and North Africa", *World Development*, Vol. 34, No. 2, pages 247-270.
- Heckman, JJ. (1976) "The Common Structure of Statistical Models of Truncation, Sample Selection and Limited Dependent Variables and a Simple Estimator for Such Models" in S. Berg (Ed.) "Annals of Economic and Social Measurement", *National Bureau of Economic Research*, Vol. 5, No. 4.
- Khan, M. (1990) "The Macroeconomic Effects of Fund-Supported Adjustment Programs", *IMF Staff Papers*, Vol. 37, pages 195–231.
- Marchesi, S. and E. Sirtori (2011) "Is Two Better Than One? The Effects of IMF and World Bank Interaction on Growth", *Review of International Organization*, Vol. 6, pages 287-306.
- Pop-Eleches, G. (2009) "From Economic Crisis to Reform: IMF Programs in Latin America and Eastern Europe", *Princeton University Press*.
- Putnam, R. (1988) "Diplomacy and Domestic Politics: The Logic of Two Level Games", *International Organization*, Vol. 42, No. 3, pages 427-460.
- Przeworski, A. and JR. Vreeland (2000) "The effect of IMF programs on economic growth", Journal of Development Economics, Vol. 62, pages 385–421.
- Steinwand, M. and R. Stone (2008) "The International Monetary Fund: A Review of the Recent Evidence", *Review of International Organizations*, Vol. 3, pages 123-149.
- Stone, RW. (2004) "The Political Economy of IMF Lending in Africa", American Political Science Review, Vol. 98, No. 4, pages 577-591, November.
- Stubbs, T., B. Reinsberg, A. Kentikelenis and L. King (2018) "How to Evaluate the Effects of IMF Conditionality. An Extension of Quantitative Approaches and An Empirical Application to Public Spending", *The Review of International Organizations*, Vol. 15.
- Sturm, JE., H. Berger and J. De Haan (2005) "Which Variables Explain Decisions on IMF Credit? An Extreme Bounds Analysis", *Economics and Politics*, Vol. 17, No. 2, July.
- Thacker, SC. (1999) "The High Politics of IMF Lending", *World Politics*, Vol. 51, No. 1, pages 38-75, October.
- Vaubel, R. (1996) "The Political Economy of International Money and Finance" in Forrest, C. and GE. Wood (Eds.) "Monetary Economics in the 1990s", *Palgrave Macmillan*.
- Vaubel, R. (1986) "A Public Choice Approach to International Organization", *Public Choice*, Vol. 51, No. 1, pages 39-57.
- Vreeland, JR. (2003) "The IMF and Economic Development", Cambridge University Press.
- Youssef, J. and C. Zaki (2019) "Between Stabilization and Allocation in the MENA Region: Are Competition Laws Helping?", *ERF Working Paper Series* No. 1319.

Annex 1	l
---------	---

	The evolution of IMF lending	g facilities
Broad Category	Lending Facility	Definition/Main Objective
	Stand-By Arrangement (SBA)	It was designed in 1952 and it aims at responding to countries' external financing needs and supporting their adjustment policies with short-term financing. It is used to solve short term balance of payments problems. It usually lasts up to one or two years.
Non- concessional	Extended Fund Facility (EFF)	Recognizing that some balance of payments problems would require longer programs, the Fund introduced the EFF in 1974. It is used to provide assistance to countries experiencing medium term balance of payments problems because of structural weaknesses that require time to be addressed. It provides support for comprehensive programs including the policies needed to correct structural imbalances over an extended period. It usually lasts up to three years.
Low-Income Countries/ Concessional -	Enhanced Structural Adjustment Facility (ESAF) and Structural Adjustment Facility (SAF)	SBA and EFF are not supposed to cover low-income countries. To that effect, the Fund established the SAF in 1986 and the ESAF in 1987 in order to provide low-interest loans to poor countries. Hence, ESAF and SAF are considered as concessional loans whereas EFF and SBA are considered as non-concessional. The interest rate for SAF and ESAF is 0.5 percent with a five-year grace period followed by repayments to be paid over a period of five to ten years.
	Poverty Reduction and Growth Facility (PRGF)	Following the East Asian crisis, in November 1999, the IMF terminated its Enhanced Structural Adjustment Facility (ESAF) and replaced it with the PRGF as a new lending facility for low-income countries.
	Extended Credit Facility (ECF)	It is the corresponding EFF for low income countries. It replaced the PRGF as the main tool for addressing balance of payments problems.
	Exogenous Shocks Facility (ESF)	It was established in 2008. It provided concessional financing to Poverty Reduction and Growth Trust (PRGT)-eligible countries facing balance of payments needs caused by sudden and exogenous shocks. It was replaced later by the SCF that became effective in January 2010.
	Standby Credit Facility (SCF)	It was created to provide support to low income countries with short-term balance of payments needs, similar to SBAs, with the possibility of using it on a precautionary basis.
Precautionary	Flexible Credit Line (FCL)	It was established in 2009 and it is designed for crisis-prevention and crisis-mitigation lending for countries with strong policy frameworks and track records in economic performance. It gives the country the flexibility to draw on the credit line at any time within a prespecified period, or to treat it as a precautionary

Table 1. The evolution of IMF lending facilities

Broad Category	Lending Facility	Definition/Main Objective
0 v		instrument. To date, five countries have used the FCL, namely Chile, Colombia, Mexico, Peru and Poland.
	Precautionary and Liquidity Line (PLL)	It was introduced in 2011. It provides financing to meet actual or potential balance of payments needs of countries with sound policies and that may have some remaining vulnerabilities. It combines a qualification process (similar to that for the FCL but with a lower bar) with ex-post conditionality. To date, only two countries have used the PLL, namely Macedonia and Morocco.
Rapid	Rapid Financing Instrument (RFI)	It was introduced in 2011 and it provides rapid and low-access financial assistance to member countries facing an urgent balance of payments need, without the need to have a full-fledged program in place. It is available to all member countries, but its similar concessional version is the RCF.
Rapid and Low-income Countries/ Concessional	Rapid Credit Facility (RCF)	It provides low-access, rapid, and concessional financial assistance to low-income countries facing an urgent balance of payments need, without ex post conditionality. It can provide support in a wide variety of circumstances, including shocks, natural disasters, and emergencies resulting from fragility.
Non- Financial	Policy Coordination Instrument (PCI)	It is a non-financial tool that is available to all IMF members that do not need Fund financial resources at the time of approval. It is designed for countries seeking to demonstrate commitment to a reform agenda or to unlock financing from other official creditors or private investors.
Non- Financial and Low-Income Countries	Policy Support Instrument (PSI)	It is a non-financial instrument that provides policy support and signals for mature stabilizers whenever countries have attained external and domestic macroeconomic stability such that they no longer needed continuous Fund financial assistance. The PSI is available to all PRGT-eligible countries that have no current or prospective balance of payments need requiring any significant macroeconomic policy adjustment, but that may still benefit from structural reforms. The PSI can expedite access to the SCF if needed.

Source: Compiled by the authors from the IMF website (factsheets on different loans); Bird and Rowlands, 2017; Bal Gunduz et al., 2013; Barro and Lee (2005)

Annex 2

List of Countries¹⁰

Afghanistan	Djibouti	Lao PDR	Samoa
Albania	Dominica	Latvia	São Tomé and Principe
Algeria	Dominican Republic	Lebanon	Senegal
American Samoa	Ecuador	Lesotho	Serbia
Angola	Egypt, Arab Rep.	Liberia	Seychelles
Antigua and Barbuda	El Salvador	Libya	Sierra Leone
Argentina	Equatorial Guinea	Lithuania	Slovak Republic
Armenia	Eritrea	Madagascar	Solomon Islands
Azerbaijan	Estonia	Malawi	Somalia
Bangladesh	Eswatini	Malaysia	South Africa
Barbados	Ethiopia	Maldives	South Sudan
Belarus	Fiji	Mali	Sri Lanka
Belize	Gabon	Marshall Islands	St. Kitts and Nevis
Benin	Gambia, The	Mauritania	St. Lucia
Bhutan	Georgia	Mexico	St. Vincent and the Grenadines
Bolivia	Ghana	Micronesia, Fed. Sts.	Sudan
Bosnia and Herzegovina	Greece	Moldova	Suriname
Botswana	Grenada	Mongolia	Syrian Arab Republic
Brazil	Guatemala	Montenegro	Tajikistan
Bulgaria	Guinea	Morocco	Tanzania
Burkina Faso	Guinea-Bissau	Mozambique	Thailand
Burundi	Guyana	Myanmar	Timor-Leste
Cabo Verde	Haiti	Namibia	Togo
Cambodia	Honduras	Nepal	Tonga
Cameroon	Hungary	Nicaragua	Tunisia
Central African Republic	Iceland	Niger	Turkey
Chad	India	Nigeria	Turkmenistan
Chile	Indonesia	North Macedonia	Tuvalu
China	Iran, Islamic Rep.	Pakistan	Uganda
Colombia	Iraq	Panama	Ukraine
Comoros	Ireland	Papua New Guinea	Uruguay
Congo, Dem. Rep.	Jamaica	Paraguay	Uzbekistan
Congo, Rep.	Jordan	Peru	Vanuatu
Costa Rica	Kazakhstan	Philippines	Venezuela, RB
Côte d'Ivoire	Kenya	Poland	Vietnam
Croatia	Kiribati	Portugal	West Bank and Gaza
Cuba	Korea, Dem. People's Rep.	Romania	Yemen, Rep.
Cyprus	Kosovo	Russian Federation	Zambia
Czech Republic	Kyrgyz Republic	Rwanda	Zimbabwe

¹⁰ This group of countries include all low- and middle-income countries according to the World Bank classification and all countries which were enrolled in any IMF agreement over the period 1993 till 2019.

Annex 3

		eomponento	2001100	-		
		EFF			PRGF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.120***	0.0459***	0.166***	0.121***	0.0429***	0.164***
	(0.0191)	(0.00471)	(0.0198)	(0.0192)	(0.00470)	(0.0199)
Schooling	0.00679	-0.00839***	-0.00160	0.00352	-0.0116***	-0.00812
	(0.0110)	(0.00272)	(0.0114)	(0.0109)	(0.00266)	(0.0113)
Nat. Rent	0.224***	0.0385***	0.263***	0.213***	0.0373***	0.250***
	(0.0343)	(0.00847)	(0.0355)	(0.0340)	(0.00831)	(0.0352)
Arab. Land	-0.0969*	0.0506***	-0.0463	-0.0921*	0.0546***	-0.0375
	(0.0502)	(0.0124)	(0.0519)	(0.0502)	(0.0123)	(0.0519)
Loan	0.291***	-0.0253	0.266***	0.533***	-0.187***	0.346**
	(0.0814)	(0.0201)	(0.0841)	(0.168)	(0.0410)	(0.174)
Constant	-2.194*	2.306***	0.112	-4.790***	3.359***	-1.431
	(1.269)	(0.313)	(1.312)	(1.603)	(0.392)	(1.660)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.070	0.112	0.098	0.068	0.126	0.093
Number of countries	95	95	95	95	95	95
		SBA			ECF	
	Cyclical	Trend	Growth	Cyclical	Trend	Growth
Investment	0.111***	0.0467***	0.158***	0.111***	0.0472***	0.158***
	(0.0199)	(0.00487)	(0.0205)	(0.0190)	(0.00464)	(0.0196)
Schooling	-0.0103	-0.00693***	-0.0172	-0.00788	-0.00609**	-0.0140
	(0.0103)	(0.00254)	(0.0107)	(0.0101)	(0.00247)	(0.0104)
Nat. Rent	0.204***	0.0403***	0.245***	0.189***	0.0341***	0.224***
	(0.0341)	(0.00836)	(0.0352)	(0.0347)	(0.00849)	(0.0358)
Arab. Land	-0.0746	0.0487***	-0.0259	-0.0742	0.0493***	-0.0249
	(0.0503)	(0.0124)	(0.0520)	(0.0500)	(0.0122)	(0.0516)
Loan	0.0185	-0.000518	0.0180	-0.177**	-0.0751***	-0.252***
	(0.117)	(0.0288)	(0.121)	(0.0842)	(0.0206)	(0.0868)
Constant	-1.757	2.261***	0.504	-2.467*	1.910***	-0.556
	(1.446)	(0.355)	(1.494)	(1.323)	(0.323)	(1.364)
Observations	1,237	1,237	1,237	1,237	1,237	1,237
R-squared	0.060	0.110	0.090	0.063	0.121	0.097
Number of countries	95	95	95	95	95	95

Effect of IMF Loans on Growth Components - BW Filter

Notes:

-

We include a fixed effect for each country in order to account for time-invariant unobservable country-specific factors.

Standard errors in parentheses *** p<0.01, ** p<0.05, * p<0.1





Figure 1. IMF Lending Intensity, 1992-2020

Source: Constructed by the authors based on data from the IMF Monitoring of Fund Arrangements (MONA) Database

Notes:

- i. The intensity of IMF lending represents the total amount of IMF lending a country has received over 1992-2020 (in million SDR).
- ii. The darker the shade of blue the higher is the amount of total IMF lending for the country over 1992-2020.